Author's response to reviews

Title: Cyclin A1 promoter hypermethylation in human papillomavirus-associated cervical cancer

Authors:

Nakarin Kitkumthorn (Art_khuan@hotmail.com)
Pattamawadee Yanatassanej (Nuchjoe@hotmail.com)
Sorapop Kiatpongsan (Ksorapop@yahoo.com)
Chureerat Phokaew (Hoshleys@hotmail.com)
Prasert Trivijitsilp (Tprasert@chula.ac.th)
Wichai Termrungruanglert (Wichaiterm@hotmail.com)
Damrong Tresukosol (Dtresukosol@hotmail.com)
Surang Triratanachat (Surangtrirattanachat@hotmail.com)
Somchai Niruthisard (Ndsomchai@hotmail.com)
Apiwat Mutirangura (mapiwat@chula.ac.th)

Version: 5 Date: 27 February 2006

Author's response to reviews:

Thank you very much for publishing this article. All changes include

Short title - removed from the title page.
Keywords section - removed the keywords section from the manuscript.
Affiliations - Each affiliations were written in full